Author's response to reviews

Title: Rare Causes of Scoliosis and Spine Deformity: Experience and Particular Features

Authors:

Konstantinos CH. Soultanis (ksoultanis@otenet.gr)
Alexandros Payatakes (ksoultanis@otenet.gr)
Vasilios Chouliaras (vhoulid@yahoo.gr)
Georgios Mandellos (ksoultanis@otenet.gr)
Panayotis N. Soucacos (soukakos@panafonet.com)

Version: 3 Date: 25 July 2007

Author's response to reviews:

Dear Sirs,

I would like to inform you that I prepared a 2nd revised paper as it was requested. In this revised paper the following changes are done:

1) A reference is added at table 1.

2) In table 2 the order of patients is changed according SRS classification. One patient with arthrogryposis is added, while two patients (muscle fiber type disproportion, Pelizaeus-Merzbacher leucodystrophy) were deleted. Total number is 13 patients.

3) The order of cases presentation in text is changed according table 2.

4) Follow up for each patient is provided.

5) Figures and legends are placed in text to fit each patient.

6) Minor language corrections were done.

Thank you for the comments and the information you provided me

Yours
K. Soultanis